Author's response to reviews

Title: Translating Evidence into Policy for Cardiovascular Disease Control in India

Authors:

- Rajeev Gupta (rajeevg@satyam.net.in)
- Soneil Guptha (gupthas@gmail.com)
- Rajnish Joshi (rioshimgims@gmail.com)
- Denis Xavier (drdenisxavier@hotmail.com)

Version: 2 Date: 6 February 2011

Author's response to reviews: see over
6\textsuperscript{th} February 2011.

To
The Editor
Health Research Policy and Systems
editorial@health-policy-systems.com

Subject: Manuscript number MS: 1520122847548924

Dear Sir:

Thanks for your favourable comments on our manuscript entitled “Translating evidence into policy for cardiovascular disease control in India”. We have made changes as suggested by the reviewers and the answers to specific comments are enclosed.

The article has been reformatted according to the journal style. We hope that you would find the article suitable for publication.

With best wishes.

Yours faithfully,

Rajeev Gupta, MD PhD
Manuscript number MS: 1520122847548924

Original article: Translating evidence into policy for cardiovascular disease control in India.

Answers to referees’ comments:

Referee 1:

Thanks or the favourable comments.

1. The references regarding mortality data from regional studies in Tamilnadu and Andhra Pradesh have been added (numbers 8 and 9).

2. Figure 2 has been totally reformatted and streamlined.

3. Spelling mistakes have been corrected.

Referee 2:

Thanks for your favorable comments.